Author's response to reviews

Title: Sex partnerships, health, and social risks of young men leaving jail: analyzing data from a randomized controlled trial

Authors:

Megha Ramaswamy (mramaswamy@kumc.edu)
Nicholas Freudenberg (nfreuden@hunter.cuny.edu)

Version: 4 Date: 10 November 2010

Author's response to reviews:

Dear Editors of BMC Public Health,

Thank you for your acceptance, in principle, of the manuscript MS: 1313624437933023, Sex partnerships, health, and social risks of young men leaving jail: analyzing data from a randomized controlled trial.

We have made the following changes to the manuscript per your editorial feedback:

1. We cited tables in numerical order in the text.
2. We removed all vertical lines from the tables.
3. We proofread the entire manuscript to ensure that there were no typos.

Thank you very much for all of your help.

All the best,

Megha Ramaswamy & Nick Freudenberg